Author’s response to reviews

Title: Effect of vitamin E (Tri E(R)) on antioxidant enzymes and DNA damage in rats following eight weeks exercise

Authors:

Noor Aini Abdul Hamid (jurishi@hotmail.com)
Mohd Hasrul A (hasrul@medic.ukm.my)
Rusdiah Ruzanna J (rusdiah@medic.ukm.my)
Ibrahim Abdel Aziz Ibrahim (jjurishi@yahoo.com)
Prasamit Saurav Baruah (prasamit@cybermed.edu.my)
Musalmah Mazlan (musalmah@medic.ukm.my)
Yasmin Anum Mohd Yusof (anum@medic.ukm.my)
Wan Zurinah Wan Ngah (zurina@medic.ukm.my)

Version: 5 Date: 20 April 2011

Author’s response to reviews: see over
Submission of Manuscript

Title: **Effect of vitamin E (Tri E®) on oxidative stress and DNA damage in rats following eight weeks exercise**

We wish to submit the above manuscript for consideration to be published in the BMC Nutrition Journal.

We certified that we participated sufficiently in the intellectual content, conception and design of this work or the analysis and interpretation of the data, as well as the writing manuscript. Neither this manuscript nor one with substantially similar content under our authorship has been published or is being considered for publication elsewhere.

This article has been read and approved by all the authors and the authors are Noor Aini Abd Hamid, Mohd Hasrul A, Rusdiah Ruzanna J, Ibrahim Abdel Aziz Ibrahim, Prasamit SB, Musalmah M, Yasmin MY and Wan Zurinah WN. We can be reached by telephone: 603-83137120, email: jurishi@hotmail.com or aini@cybermed.edu.my or by the above address.

There is no financial interest, direct or indirect, that exists or may be perceived to exist for individual contributors in connection with the content of this article.

We hope that you could kindly consider this paper for publication in your reputed Journal.

Thank You.

Yours Sincerely,

**Prof. Dr. Noor Aini Abd Hamid**